Author's response to reviews

Title: Combined inhibition of the cell cycle related proteins Wee1 and Chk1/2 induces synergistic anti-cancer effect in melanoma

Authors:

Gry I Magnussen (gry.magnussen@rr-research.no)
Elisabeth Emilsen (eee@ous-hf.no)
Karianne G Fleten (Karianne.Giller.Fleten@rr-research.no)
Birgit Engesæter (Birgit.Engesater@rr-research.no)
Viola Nähse-Kumpf (Viola.Naehse@rr-research.no)
Roar Fjær (Roar.Fjar@medisin.uio.no)
Ana Slipicevic (Ana.Slipicevic@rr-research.no)
Vivi A Flørenes (vivi.ann.florenes@radiumhospitalet.no)

Version: 2 Date: 27 November 2014

Author's response to reviews:

We have made the following alterations to the manuscript:

1. Title page
   Email addresses of all authors have been included

2. Line numbering
   Line numbering and page numbers have been included